
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Mon Aug 19 16:17:01 GMT 2002
Year: 02  Doy: 231
Observer: yasukawa
WEATHER COMMENT: Mon Aug 19 16:17:04 GMT 2002
Cool, in broken cirrostratus, varying thickness, light south wind.
**PSPT PROBLEM**: Mon Aug 19 16:26:37 GMT 2002
Still getting messages "no space left on device" on xterm window referring
to data transfer attempts overnight.
Mon Aug 19 16:34:06 GMT 2002    CHIP     Start Patrol
Mon Aug 19 16:34:17 GMT 2002    PICS     Start Patrol
**MLSO PROBLEM**: Mon Aug 19 16:26:28 GMT 2002
**MLSO PROBLEM**: Mon Aug 19 17:03:14 GMT 2002
hoike is still not updating today.

No tape status message on email today, although I did get an email
indicating that dlt tape is full and there was no log entry made to
indicate that tape was changed.

Alerted trouble to look into the problems occurring over the weekend in
case Alice (Darryl sent emails to Alice during the weekend to report
the problems) is not around, in transit to MLSO.
Mon Aug 19 18:04:13 GMT 2002    PICS     Flat
Mon Aug 19 18:07:03 GMT 2002    PICS     End Flat
Mon Aug 19 19:01:33 GMT 2002    CHIP     LSD
Mon Aug 19 19:03:13 GMT 2002    CHIP     End LSD
Mon Aug 19 19:03:19 GMT 2002    CHIP     BiasLSD
Mon Aug 19 19:04:09 GMT 2002    CHIP     End BiasLSD
Mon Aug 19 19:04:15 GMT 2002    CHIP     Bias
Mon Aug 19 19:05:06 GMT 2002    CHIP     End Bias
Mon Aug 19 19:27:39 GMT 2002    CHIP     End Patrol
Mon Aug 19 19:27:50 GMT 2002    PICS     End Patrol
COMMENT: Mon Aug 19 19:27:48 GMT 2002
Stopping to reconfigure dome shutter.
Mon Aug 19 19:33:52 GMT 2002    MKIV    Start Patrol
WEATHER COMMENT: Mon Aug 19 19:34:09 GMT 2002
Clear hole.
Mon Aug 19 19:34:11 GMT 2002    CHIP     Start Patrol
Mon Aug 19 19:34:17 GMT 2002    PICS     Start Patrol
A NEW TAPE HAS BEEN PUT INTO KAIEE DLT DRIVE, Mon Aug 19 19:45:49 GMT 2002
PSPT COMMENT: Mon Aug 19 19:49:42 GMT 2002
Starting daily obs.
WEATHER COMMENT: Mon Aug 19 19:53:50 GMT 2002
Back in cirrostratus.
Mon Aug 19 19:57:45 GMT 2002    MKIV    End Patrol
**MLSO PROBLEM**: Mon Aug 19 21:51:42 GMT 2002



Alice and Bunny resolved the Kaiee memory problem and the fallout 
it caused.  Alice reports that there is still the nahenahe mail problem
where the logs are not being sent to Boulder.  She has referred that 
problem to CSMT for resolution.
Mon Aug 19 22:03:00 GMT 2002    CHIP     End Patrol
Mon Aug 19 22:15:14 GMT 2002    PICS     End Patrol
PICS COMMENT: Mon Aug 19 22:17:07 GMT 2002
PICS did not "hear" my request to end patrol.  Had to start Video and use
Abort to get its attention.
Mon Aug 19 22:18:26 GMT 2002 
        MkIV    
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